Author's response to reviews

Title: An Empirical Study to Determine Factors That Motivate and Limit The Implementation Of ICT In Healthcare Environments

Authors:

Raj Gururajan (<u>gururaja@usq.edu.au</u>) Abdul Hafeez-Baig (<u>abdulhb@usq.edu.au</u>)

Version: 2 Date: 19 August 2013

Author's response to reviews:

1. we have reformatted the article as per the email received and making a revised submission